Author's response to reviews

Title: Biomechanical Comparison of Lumbar Spine Instability between Laminectomy and Bilateral Laminotomy for Spinal Stenosis Syndrome- An Experimental Study in Porcine Model

Authors:

Ching-Lung Tai (taicl@mail.cgu.edu.tw)
Pang-Hsing Hsieh (hsiehph@adm.cgmh.org.tw)
Weng-Pin Chen (wpchen@cycu.edu.tw)
Lih-Huei Chen (lhchen2132@adm.cgmh.org.tw)
Wen-Jer Chen (chenwenj@adm.cgmh.org.tw)
Po-Liang Lai (ortholab@adm.cgmh.org.tw)

Version: 5 Date: 19 May 2008

Author's response to reviews: see over